Author's response to reviews

Title: Can cancer patients assess the influence of pain on functions? A randomised, controlled study of the pain interference items in the Brief Pain Inventory

Authors:

- Guri Stenseth (guris@stud.ntnu.no)
- Mari Bjornes (maribjor@stud.ntnu.no)
- Stein Kaasa (stein.kaasa@ntnu.no)
- Pal Klepstad (pal.klepstad@ntnu.no)

Version: 4 Date: 8 March 2007

Author's response to reviews:

Changing RESULTS to Results and 5. References to 6. References.